Author's response to reviews

Title: Is expanding HPV vaccination programs to include school-aged boys likely to be value-for-money? A cost-utility analysis in a country with an existing school-girl program

Authors:

- Amber L Pearson (amber.pearson@otago.ac.nz)
- Giorgi Kvizhinadze (giorgi.kvizhinadze@otago.ac.nz)
- Nick Wilson (nick.wilson@otago.ac.nz)
- Megan Smith (megan.smith@unsw.edu.au)
- Karen Canfell (k.canfell@unsw.edu.au)
- Tony Blakely (tony.blakely@otago.ac.nz)

Version: 6 Date: 11 June 2014

Author's response to reviews:

The abstract now is <350 words and includes the following sub-headings:
- Background
- Methods
- Results
- Conclusions
- Trial registration